Author's response to reviews

Title: Serial exercise testing in children, adolescents and young adults with Senning repair for transposition of the great arteries

Authors:

Roselien Buys (roselien.buys@faber.kuleuven.be)
Werner Budts (werner.budts@uzleuven.be)
Tony Reybrouck (tony.reybrouck@uzleuven.be)
Marc Gewillig (marc.gewillig@uzleuven.be)
Luc Vanhees (luc.vanhees@faber.kuleuven.be)

Version: 3 Date: 7 May 2012

Author's response to reviews:

Dear Editor,

On behalf of the co-authors and myself, I would like to submit the revised manuscript entitled ‘Serial exercise testing in children, adolescents and young adults with Senning repair for transposition of the great arteries', for possible publication in BMC Cardiovascular Disorders.

Information on ethics are provided in the patients paragraph of the methods section of this manuscript.

Thank you for your kind consideration.

Sincerely yours,